Author's response to reviews

Title: Conceptualizing childhood health problems using survey data: a comparison of key indicators

Authors:

Dafna E Kohen (dafna.kohen@statcan.ca)
Jamie C Brehaut (jbrehaut@ohri.ca)
Rochelle E Garner (rochelle.garner@statcan.ca)
Anton R Miller (amiller@cw.bc.ca)
Lucyna M Lach (lucy.lach@mcgill.ca)
Anne F Klassen (aklass@mcmaster.ca)
Peter L Rosenbaum (rosenbau@mcmaster.ca)

Version: 5 Date: 26 November 2007

Author's response to reviews:

Dear Editor:

Re: BMC Pediatrics 1995659993133548

Conceptualizing childhood health problems using survey data: a comparison of key indicators

Thank you for your comments. We have make the following requested changes:

- The requested formatting changes were completed (a comma was inserted between the surname and the forename initial
- The tracked changes function was turned off
- All numbers were changed into the UK numbering format

Please let us know if you have any other questions or concerns.

Sincerely,
Dafna E. Kohen